

Received September 5, 2020, accepted September 9, 2020, date of publication October 5, 2020, date of current version October 14, 2020. *Digital Object Identifier* 10.1109/ACCESS.2020.3027074

# Enabling DSRC and C-V2X Integrated Hybrid Vehicular Networks: Architecture and Protocol

## ZEESHAN HAMEED MIR<sup>®1</sup>, (Senior Member, IEEE), JAMAL TOUTOUH<sup>2</sup>, FETHI FILALI<sup>3</sup>, (Senior Member, IEEE), AND YOUNG-BAE KO<sup>®4,5</sup>

<sup>1</sup>Faculty of Computer Information Science, Higher Colleges of Technology (HCT), Fujairah, United Arab Emirates

<sup>2</sup>Computer Science and Artificial Intelligence Laboratory, Massachusetts Institute of Technology, Cambridge, MA 02139, USA

<sup>3</sup>Qatar Mobility Innovations Center (QMIC), Qatar University, Doha, Qatar

<sup>4</sup>Department of AI Convergence Network, Ajou University, Suwon 16499, South Korea
<sup>5</sup>Department of Computer Engineering, Ajou University, Suwon 16499, South Korea

Corresponding author: Young-Bae Ko (youngko@ajou.ac.kr)

This work was made possible by the National Priorities Research Program (NPRP) Grant No.: 5-1080-1-186 from the Qatar National Research Fund (a member of The Qatar Foundation). The statements made herein are solely the responsibility of the authors. This work was, in part, supported by the Civil-Military Technology Cooperation Program in Korea (19-CM-TN-07).

**ABSTRACT** Emerging Vehicle-to-Everything (V2X) applications such as Advanced Driver Assistance Systems (ADAS) and Connected and Autonomous Driving (CAD) requires an excessive amount of data by vehicular sensors, collected, processed, and exchanged in real-time. A heterogeneous wireless network is envisioned where multiple Radio Access Technologies (RATs) can coexist to cater for these and other future applications. The primary challenge in such systems is the Radio Resource Management (RRM) strategy and the RAT selection algorithm. In this article, a Hybrid Vehicular Network (HVN) architecture and protocol stack is proposed, which combines Dedicated Short-Range Communication (DSRC) technology-enabled ad hoc network and infrastructure-based Cellular V2X (C-V2X) technologies. To this end, we address the design and performance evaluation of a distributed RRM entity that manages and coordinates Radio Resources (RR) in both RATs. Central to distributed RRM are adaptive RAT selection and Vertical Handover (VHO) algorithms supported by two procedures. (1) Measurement of Quality of Service (QoS) parameters and associated criteria to select the suitable RAT according to the network conditions. (2) Dynamic communication management (DCM) via implementing RR-QoS negotiation. The simulation results show the effectiveness of the proposed architecture and protocol suite under various parameter settings and performance metrics such as the number of VHOs, packet delivery ratio, and throughput, and latency.

**INDEX TERMS** C-V2X, DSRC, hybrid vehicular networks, IEEE 802.11p, LTE, RAT selection, vertical handover (VHO).

#### I. INTRODUCTION

The emerging vehicular networking applications and use cases demand stringent Quality of Service (QoS) requirements in terms of latency, data rate, reliability, and communication range. These performance requirements are hard to meet by a single communication technology [1]. Several Radio Access Technologies (RATs) exist for vehicular networking but predominantly include two RATs. (1) The Dedicated Short-Range Communication (DSRC) technology that allows short-range, un-coordinated communication among vehicles and between vehicles and Roadside Units (RSUs), thus establishing Vehicular Ad Hoc Networks (VANETs). (2) The Cellular Vehicle-to-Everything

The associate editor coordinating the review of this manuscript and approving it for publication was Weizhi Meng<sup>(D)</sup>.

(C-V2X) [2] technology is wildly considered as a feasible alternative for providing vehicular communications because it offers superior performances in terms of throughput and lower latencies. Moreover, simplified network architecture and advanced algorithms for resource management resulted in lower cost and higher performance efficiency. Combing these two competitive standards bring immense opportunities as well as challenges to provide seamless connectivity that could not only enhance existing applications but also spur an array of new applications and services. However, a multi-RAT environment not only necessitates well-defined network architecture but also signify the need for new protocols and algorithms at each layer of the protocol stack.

In the literature, two alternatives have been discussed to combine DSRC and C-V2X using a dual-interface enabled V2X communication system. The first alternative requires

frequency sharing between the two competing technologies [3]. To reap the benefits such as interoperability between two incompatible RATs requires permitting both to coexist in the 5.85-5.925 GHz frequency range. Considering that such a proposal is at early stages, exclusive access of ITS band for vehicular safety applications, and potential negative impact of adjacent channel operation, such coexistence often leads to performance degradation [4]. The second alternative seamlessly switches between the two access technologies based on the coupling of the QoS requirement and RAT selection strategy. A DSRC-enabled direct, ad hoc communication link can be used between nearby vehicles to exchange time-critical messages. The C-V2X might be used simultaneously with DSRC to increase the reliability in cases where DSRC-enabled links fail [5]. Similarly, C-V2X can be applied effectively to distribute application data among many users within an extended area as an alternate or backup system [6]. Moreover, applications that require full internet access with higher data rates, C-V2X, can be applied as the preferred RAT. But these advantages come with the challenges of its own, for example, how to assign a particular vehicular user to the most suitable radio access network, i.e., RAT selection while lowering the signaling cost of performing Vertical Handover (VHO)? Moreover, it is also essential to efficiently manage and utilize the Radio Resources (RR) within each RAT, to provide applications with the required level of QoS.

In heterogeneous wireless networks, Radio Resource Management (RRM) functions such as RAT selection and VHO have been an active research topic. In the context of dual-interface enabled vehicular networks, extensive work is available on combining and Wireless Local Area Network (WLAN) and Wireless Wide Area Network (WWAN). For example, Olivera et al. in [7] implemented a prototype using IEEE 802.11b/g for the primary interface while 3G as the secondary interface. Similarly, bulk of research on RAT selection and VHO such as [8]-[11] and [12] proposed the integration of IEEE 802.11 based WLAN and CDMA2000, UMTS, WiMAX, or 3G/4G based WWANs. Generally, the RRM functionalities are implemented in infrastructure nodes only, i.e., integrated into the Access Point (AP), Base Station Controller (BSC), Radio Network Controller (RNC), or Evolved Node B (eNodeB). However, due to the distributed nature of vehicular networks, this approach often leads to scalability issues as the number of connections between the vehicles and RRM entities is increased [13]. The coexistence of multi-RAT and the dual-interface support is critical to the successful implementation of the hybrid communication systems herein, referred to as Hybrid Vehicular Network (HVN), which brings together the DSRC and C-V2X that operate in ad hoc and infrastructure modes, respectively. The HVN requires a distributed implementation of ad hoc domain RRM in each vehicle, which is functionally equivalent to the RRM in infrastructure-based networks. Moreover, enhanced network architecture with new protocols to support dynamic communication and resource management over HVN are highly sought-after.

In this article, a generic framework called *CellCar* is presented along with enhanced architecture and protocols for dynamic RAT selection and communication management in DSRC and the C-V2X integrated HVN. For the rest of the discussion, we will refer to current standards IEEE 802.11p [14] and the Long Term Evolution (LTE) [15] (LTE, LTE-V2x, and C-V2X are used interchangeably) for ad hoc and cellular communication technologies, respectively. However, the framework is not limited to these technologies, other evolutionary alternatives such as 802.11bd and un-supervised PC5 interface (via C-V2X sidelink Mode 4) operating and sharing the spectrum in the same 5.9 GHz frequency band could also be considered. Similarly, the 5G NR (New Radio) or the Uu interface could also allow vehicles to use the mobile cellular network for V2X communications. Because of the interoperability, coexistence, and backward compatibility requirements of these future standards [16], for the large part, the concepts presented in this article remain valid from the practical perspective.

Instead of considering DSRC and C-V2X technologies as competitors, a merger to create a hybrid network leverages the best of both. Based on the LTE Device-to-Device (D2D) platform [17], [18], the Direct communication over the PC5 interface and network communication over Uu interface and its coexistence with DSRC would result in reduced cost, improved reliability, and lower complexity. The HVN could also take advantage of reusing the already established upper layers, such as application, security, and transport layers of the protocol stack. Moreover, RSUs with DSRC and PC5 interfaces and 4G/5G small cells with the Uu interface further reduce the cost of infrastructure deployment. Finally, the coexistence of both technologies promises to meet the challenging performance requirements posed by emerging applications and use cases.

Since in DSRC, the channel access is distributed, i.e., without centralized coordination, the proposed solution translates resource management in the ad hoc domain, i.e., Primary RRM into efficient radio access. The efficiency in radio access is achieved by employing congestion control based on locally available measurements. To this end, a beaconing rate adaptation technique is employed to adapt the traffic over the shared radio channel according to the application QoS requirements. Similarly, resource management in the LTE network is controlled by Secondary RRM through preemption-based admission control. To provide coordination between Primary and Secondary RRMs, we introduce CellCar Radio Resource Management (CRRM), which allows communication systems in individual vehicles instead of the network to make the RAT selection and VHO decisions. Central to the proposed framework is an adaptive RAT selection and VHO algorithm. The RAT selection algorithm selects the most suitable technology while seamlessly maintaining the connectivity via VHO. The proposed algorithm relies on criteria based on application QoS requirements, channel condition, and network load at the DSRC and LTE networks. An Application Profiler automatically derives the QoS

requirements of an application, and initial RAT selection is performed according to the type, priority, and the traffic of the application it carries. Finally, we include details of *Dynamic Communication Management (DCM)* via implementing *RR-QoS negotiation algorithm*. The main focus of the simulation-based study is two-fold. Firstly, to provide a quantitative assessment of combining two different RATs and its potential to cater to a diverse set of QoS requirements under several parameter settings. Secondly, to show that the hybrid approach and the signaling procedures can deliver more data reliably, with lower latencies while avoiding unnecessary switching between the two RATs.

The rest of the paper is organized as follows. Section II provides an overview of the related work. The details of the proposed architecture and protocol stack are given in Section III. Section IV presents the criteria used for RAT selection in both DSRC and LTE networks. Section V provides details of the proposed RAT selection and VHO algorithms, followed by the DCM scheme in Section VI. Section VII describes the simulation scenario and performance evaluation. Finally, the paper is concluded in Section VIII.

## **II. RELATED WORK**

In recent years, the use of multi-technology enabled vehicular communication has gained widespread attention. Central to this concept are the two fundamental assumptions. (1) The communication subsystem on all or a subset of vehicles is equipped with several RATs. (2) The communication subsystem is capable of selecting and switching among different RATs by performing VHO.

Most of the research in the Heterogeneous Vehicular Network (HetVNet) [19] focused on utilizing different technologies such as Wi-Fi, WiMAX, UMTS, and 3G/4G as a standby alternative for data offloading. Such architecture usually follows Always-Best-Connectivity (ABC) [7] paradigm where the always-on and seamless connectivity is provided by consistently selecting the best RAT. Primarily, the alternative technologies act as a backup with the RAT selection algorithm is at the center of the ABC concept involving VHO. The Media Independent Handover (MIH) protocol [20] by IEEE 802.21 Working Group and Communication Access for Land Mobile (CALM) [21] is the leading standardization efforts that can be applied in a vehicular networking environment to provide optimized handover mechanism between various network technologies. There also exist several lightweight approaches as an alternative to complex standards such as MIH and CALM that consist of a considerable number of modules. For example, Olivera et al. in [7] developed a simple, transparent, and application-level handover mechanism for vehicular communications. The scheme mainly utilized Internet Protocol (IP) reachability information to switch between IEEE 802.11p and 3G interfaces. Similarly, Marquez-Barja et al. [22] proposed a VHO mechanism based on the IEEE 802.21 standard. The criteria for selecting the wireless access network considers many factors such as user preferences, price-performance based metric, and location with map layout and route itinerary information.

Initially, researchers proposed the idea of utilizing co-located hotspots or WLAN and WWAN technologies together. These studies mainly differ in two aspects, in their choice of WWAN technology and the switching mechanism, i.e., RAT selection algorithm. Hasib and Fapojuwo [8] proposed an algorithm that alternated between IEEE 802.11 and CDMA2000 network based on the service type, location, and congestion-level information. Similarly, the scheme proposed by Vegni et al. in [9] considered QoS parameters like channel resource utilization and latency to switch between Wi-Fi and UMTS network. In [10], Shafiee et al. exploited vehicular mobility profile and network characteristics to perform VHO between WLAN APs and CDMA2000 1x-EV networks. In their follow-up work, [11] assumed ad hoc communications over WLAN and WiMAX technologies. It proposed a hybrid routing protocol combining both location-based routing over WLAN-based links and topology-based paths over the WiMAX network.

Similarly, the rationale behind WiFi-3G integration given in [12] is that the data delivery process must take into account 3G budget constraints while achieving a good trade-off between delay and delivery success rate. Accordingly, Zhao *et al.* work rely on the estimated packet delivery probabilities over the ad hoc network. Only those packets are delivered through the 3G network interface with lower expectations, whereas all other packets are transported via multi-hop transmissions using Epidemic routing.

In our previous work, we suggested combining a VANET and 4G/LTE cellular networks. For such scenarios, the distinguishing factor for the VHO decision algorithm is the use of ad hoc mode rather than the traditional infrastructure model. Authors in [23]–[25] proposed and implemented a hybrid vehicular network architecture by combining IEEE 802.11p standard and 4G/LTE. The QoS-aware RAT selection and VHO algorithms took into account network load and application's requirements before switching and steering traffic between the alternative technologies while reducing the number of VHOs performed. Mir et al. considered 4G/LTE to provide infrastructure-assistance [6], [26] by steering control packets containing in-network status information over the mobile cellular network and data packet over the ad hoc network. The status information is collected at a centralized remote server, which subsequently utilizes it to decide on power adaptation and hybrid routing protocol.

In some approaches, instead of all vehicles, only a few static or mobile gateway or cluster-head vehicles are required to be equipped with dual-interface enabled communication systems. All other vehicles opportunistically offload vehicular data to nearby gateways or cluster-head vehicles using ad hoc networks. The gateways act as the first hop from the source vehicle and last hop towards the final destination vehicle. Typically, the system follows multi-tiered hierarchical architecture [5] in which gateways or cluster-heads collect and send aggregated data one level up in the hierarchical

organization. Central to this approach is the selection of the cluster-heads or gateways. In the context of heterogeneous networks, the cellular network infrastructure is used mainly in two ways. In the first approach, proposed in [27], the base station or a remote server assists in creating and maintaining the cluster organization (or clustering) in a centralized fashion. The second approach assumes that the grouping is already in place and refers to selecting an optimal number of mobile gateways [28] in a distributed manner. The mobile gateways act as the relaying element between the vehicles and the cellular network infrastructure. To this end, several gateway or cluster-head selection algorithms have been proposed that mainly differ in their choice of selection criteria. Abderrahim et al. in [29] utilized speed, Received Signal Strength (RSS), and link stability information for gateway selection in a VANET-UMTS (3G) integrated vehicular network.

Similarly, Rajarajan *et al.* in [30] proposed a multi-criteria gateway selection algorithm in a VANET-LTE enabled vehicular network by considering the DSRC transmission rate, LTE channel quality, and relative distance metrics. Finally, in [31], [32], Zhioua *et al.* used QoS traffic classes and link connectivity information for electing the gateway vehicle. More recently, Ion *et al.* proposed a timer-based cluster-head selection algorithm, where the timer value depends on the number of neighbors and LTE channel quality [33].

Compared with the related work, the main contributions and novelties of this article are given below.

- 1) Unlike most of the earlier work where the RRM functionalities are implemented in their respective infrastructure nodes only, we proposed a two-tiered RRM strategy. The distributed CRRM module at the upper-tier manages the Primary RRM, i.e., DSRC RAT, and provides coordination with radio resource allocation at the lower-tier in the Secondary RRM, i.e., C-V2X RAT.
- 2) An adaptive RAT selection algorithm that takes in to account QoS metrics such as channel occupancy level in the Primary RAT and network load in the Secondary RAT. To select the most appropriate RAT according to network conditions, it also includes a network-assisted, but the vehicle-initiated VHO mechanism, which ensures seamless connectivity in the best possible way.
- 3) A dynamic communication management module performs QoS negotiation of critical communication parameters against the application's QoS requirements to reduce unnecessary switching between RATs significantly. Towards this end, the DCM module implements beaconing rate adaptation and admission control in the Primary RRM and the Secondary RRM.
- 4) A suite of protocols was developed and tested that allows a unified communication system by interweaving CRRM, RAT selection with VHO, and DCM algorithms. The CellCar framework provides performance

improvements in a heterogeneous vehicular networking environment using DSRC and C-V2X technologies.

## III. SYSTEM ARCHITECTURE AND PROTOCOL STACK

## A. COMPONENTS OF THE PROPOSED ARCHITECTURE

This section provides details on the dual-interface enabled V2X communication system architecture and protocol stack. A CS can perform V2X communication based on two established links. The DSRC-enabled direct link to exchange the time-critical messages between nearby vehicles. Based on IEEE 802.11p, the direct link uses radio resources at the 5.9 GHz frequency band. Alternatively, use the LTE link, where messages are exchanged reliably over the more extensive range by passing through the cellular infrastructure. The 4th Generation (4G) mobile radio system LTE offers a highly flexible network located in the 800 MHz, 1800 MHz, and 2600 MHz band with a channel width between 1.4 and 20 MHz.

- 1) CellCar Station (CS): is a vehicle running CellCar. It comprises an application unit (CCApp) where the messages are generated and passed to an onboard unit (OBU). The OBU is connected to dual-interface radio transceivers embedded with DSRC and C-V2X RATs. The CellCar system supports the Global Positioning System (GPS) component for providing location information. The OBU also implements optimized protocol architecture design as suggested by relevant standardization bodies, e.g., Wireless Access in Vehicular Environment (WAVE), Communication Access for Land Mobile (CALM) [21], Car-2-Car Communication Consortium (C2C-CC) [34]. The optimization is achieved by integrating new methods, protocols, and services in the data plane layers and the associated management *plane* of the protocol stack.
- 2) CellCar Information Center (CIC): is a central entity that runs several back-end services such as location management. The CIC is also responsible for sending application data from one CS to one or several CS in a specific geographical Area of Interest (AOI).
- 3) CellCar Application (CCApp): comprises all the application types and defines the format of application-level data messages exchanged among CSs and between CS and CIC. As given in the ETSI TC ITS [35] standard, CCApp can generate one of the two types of messages. (1) Cooperative Awareness Message (CAM) that are generated periodically and (2) Distributed Environmental Notification Message DENM) that are triggered due to the occurrence of an Event of Interest (EOI). Intelligent mechanisms facilitate the handling of application data at the CIC. It includes selecting the destination geographical areas that require messages to carry the destination area (i.e., message type GA).
- 4) **DSRC-enabled Ad hoc Network:** In the DSRCenabled network, the CAM/DENM are exchanged in a distributed fashion using single-hop broadcasts



**FIGURE 1.** CellCar system architecture with five major components, CellCar Application (CCApp), CellCar Station (CS), CellCar Information Center (CIC), ad hoc and mobile cellular networks.

from each CS to its immediate neighborhood. Alternatively, position-based multi-hop communication is performed to reach more distant destination CSs within the AOI disseminating through specific geographical forwarding region. The GeoNetworking [36] carries out the latter technique layer, which performs position-based geographical forwarding of the messages. From the protocol architecture perspective, two configurations exist. A simple transport protocol stacked over the GeoNetworking layer, which transmits messages using short-range wireless communication standard for vehicular networks. Alternatively, messages can be carried by the GeoNetworking layer with Transport Control Protocol (TCP)/IP or User Datagram Protocol (UDP)/IP application on top.

5) Mobile Cellular Network: The cellular network architecture comprises a radio access network (E-UTRAN), which takes care of all radio control and management functionalities and interaction between the User Equipment (UE) and Evolved NodeB (eNodeB). The eNodeB is directly connected with the Evolved Packet Core (EPC), which supports mobility management, QoS handling, and interoperability with legacy 3GPP and Non-3GPP access technologies. LTE support IP-based data transmission; therefore, messages can be encapsulated and transported as an IP packet payload with TCP/IP or UDP/IP applications running directly on top of the wireless access network. As another option is to run applications on top of the GeoNetworking layer that, in turn, utilizes the IP protocol to transmit the datagrams. In the latter case, the dissemination of application data over a cellular network requires the presence of CIC like centralized entity. The CIS server receives data transmitted by the CSs and further relays it towards the destination CSs within the AOI.

As Fig. 1 shows, CSs can communicate with other CSs either directly through DSRC based ad hoc communication

or via infrastructure-assistance in the form of CIC. The CIC is either interconnected to a network element within eNodeB or EPC or accessible through the Internet. From the data dissemination perspective, CS sends the vehicle's movement and application data to the CIC server(s) and immediate neighboring CSs. The CIC's back-end server implements services such as storage and management of location information and forwarding and aggregation of messages. Furthermore, LTE supports both CS and UE background traffic.

## **B. PROTOCOL STACK DESIGN**

The protocol design is based on a convergent architecture, which includes an optimized dual-protocol stack combining DSRC and GeoNetworking. The protocol stack comprises an application, facilities, transport and networking, and access layers. A WAVE Management Entity (WME) manages the upper and the lower layers of the protocol stack, which constitutes the *management plane*. Simultaneously, the layers that operate on the messages themselves are known as the *data plane*. Fig. 2 shows the detailed protocol architecture design. The WME registers priority, data rate, and power for different applications.

- Application Layer: The proposed protocol design architecture supports four types of vehicular applications, i.e., active road-safety, Enhanced-V2X (e-V2X) [37], traffic efficiency, and infotainment. The application layer assigns each application a unique application identifier. Applications are classified according to four basic types, and corresponding priorities and initial RAT preferences are determined. Finally, their functional and QoS requirements are registered in several communication performance metrics, such as the beaconing rate.
- 2) Facilities Layer: supports application, information, and communication functionalities. Among others, the essential supported features are providing location information, Local Dynamic Map (LDM), and



FIGURE 2. Protocol stack design for the CellCar system.

CAM/DENM management. Additionally, build on top of existing functionalities, in the proposed framework, the facilities layer contains support for Application Profiler and Bootstrap RAT Selection. More details are given in Section V.

- 3) Transport and Network Layers: The communications among CS is performed in the DSRC access network over *Primary RAT* interface. The position-based forwarding strategy or GeoNetworking is employed for application data dissemination. Whereas transmission between CS and CIC server via the Internet is carried out in LTE access networks using *Secondary RAT* interface. Two alternatives exist, first end-to-end with application data encapsulated in the IP packet payload and IP over GeoNetworking. The networking protocol is connected to a simple, dedicated transport protocol for vehicular communication or the TCP or UDP. The networking layer also maintains the neighbor and routing tables containing locations of the direct neighboring CS.
- 4) Access Layer: is responsible for the Physical and Data Link Layers of each communication interface. The access layer monitors the communication interface and provides information on several status indicators, such as busy channel rate, RSS, and frame transmission statistics. Moreover, the access layer is also capable of manipulating the parameters of the specific communication interface. In the context of the hybrid vehicular network, the connectivity for CS is provided by two access networks technologies, including DSRC based system and LTE network, also Primary RAT interface and Secondary RAT interface, respectively.

5) Management Layer: Generally, the management layer manages different aspects of the service advertisement, communication service, and networking management. The two functionalities that the proposed framework extends significantly are the congestion control and the mapping of application data flows on the available communication interfaces. The *CellCar Radio Resource Management (CRRM)* primarily describes the component that implements effective communication and resource management in a multi-technology vehicular network.

Central to the multi-technology enabled CS is the CRRM, which follows a two-tiered model where an individual RMM manages each access network, i.e., Primary RAT and Secondary RAT interfaces are controlled by Primary RRM and Secondary RRM, respectively. The CRRM not only coordinates with each RRM but also communicates with other CRRM entities. Each RRM is responsible for implementing strategies like power control, packet scheduling, congestion control, admission control, and handover control, etc. for their respective access networks. Due to higher interaction between CRRM and RRMs and dynamic RRM handling, CRRM is implemented in a distributed manner. Therefore, RAT selection, VHO, and DCM for the Primary RAT are left for the CS to handle and not to a centralized entity. One of the benefits of moving CRRM functions to the CS is that all necessary measurements, such as Channel Waiting Time (CWT), Queue Length (QL), originate at the CS terminal. Moreover, the CS also contains the application interface, which provides information on



FIGURE 3. Two-tiered topology for the CellCar Radio Resource Management (CRRM) entity.

the application's QoS requirements. Fig. 3 shows the high-level abstraction for the CRRM and its two-tiered topology.

From the functional perspective, the coordination between CRRMs and individual RRMs is carried out by mean of information reporting and decision support functions [13]. (1) Information Reporting: The information reporting sub-module is responsible for monitoring and exchanging local measurements and QoS information associated with both Primary RRM and Secondary RRM and among CRRM entities. The sharing of information can be done either periodically or as the result of a particular condition. (2) Decision Support: The shared information is acted upon and conveyed by the decision support sub-module where RRMs make decisions regarding radio resource utilization either independently or as per instructed by the CRRM, which work in close interaction with other CRRM entities.

#### **IV. SELECTION CRITERIA FOR DSRC AND LTE NETWORKS**

The proposed RAT selection and VHO algorithms utilize in-network information such as channel occupancy level and cellular network load, monitored and coordinated by the Primary RRM and Secondary RRM entities. In the former case, the Primary RRM only observes the local (one-hop) neighborhood area, whereas the Secondary RRM coordinates available resources at the single macro-cell site. To this end, the critical parameters considered for both the RAT selection and VHO algorithms are given in the following section.

**A.** CHANNEL OCCUPANCY AS DSRC SELECTION CRITERIA In this article, the *Channel Waiting Time* (*CWT*) and *Queue Length* (QL) are considered to measure the congestion in terms of *Channel Occupancy Level (COL)* in the vehicular ad hoc network. The COL estimates based on the CWT, and QL is obtained from the Access layer [38].

## 1) CALCULATING CHANNEL WAITING TIME (CWT)

The frequency channel layout of a 5.9 GHz DSRC consists of seven channels, each with the bandwidth of 10 MHz. One channel is reserved as a control channel (CCH), four channels are allocated as service channels (SCHs), and two are for future use. The road-safety and control messages are transmitted over CCH. In contrast, SCHs carries messages for traffic efficiency and infotainment applications during the fixed CCH Interval (CCHI) and SCH Interval (SCHI), respectively. The DSRC standard utilizes Enhanced Distributed Channel Access (EDCA) [39] method to enable QoS for packet transmission. EDCA uses four queues to differentiate among the different levels of priorities, also called Access Categories (AC). Each AC, in turn, is dedicated to another type of data traffic. For example, we assume that the higher priority queue, i.e., AC0 relates to road-safety applications and AC1 concerns with the Enhanced-V2X (e-V2X) [37] use cases such as teleoperated remote automated driving with stringent latency requirements. Whereas the other two access categories AC2 and AC3, hold traffic for delay-tolerant traffic efficiency and comfort applications. In EDCA, the priority is established by implementing several delay parameters such as Arbitration Inter Frame Space Number (AIFSN) and Contention Window (CW) with the basic principle, which states that the smaller these parameter values higher the priority, thus earlier the chances of transmissions. For further information on delay parameters and their values corresponding to each AC, refer [39]. Consider the following parameter settings.

- Maximum CCHI and SCHI:  $T_{max} = 50$  ms.
- Number of queues: q = 4.

- Medium busy time of m<sup>th</sup> message at the i<sup>th</sup> queue as indicated by the Channel Clear Assessment CCA) module: T<sup>i</sup><sub>m</sub> where i ∈ 0, 1, 2, 3.
- aSlotTime =  $32 \ \mu s$ .
- Short Inter-Frame Space (SIFS) =  $13 \ \mu s$ .

The following equation gives the mean back-off duration.

$$T^{i}_{backoff} = 0.5 \times aCW^{i}_{min} \times aSlotTime \tag{1}$$

The AIFS duration for the  $i^{th}$  queue with corresponding Arbitration Inter Frame Space Number (AIFSN) is given as follows.

$$T^{i}_{AIFS} = SIFS + AIFSN^{i} \times aSlotTime$$
(2)

Then, the average waiting time for  $m^{th}$  messages at the  $i^{th}$  queue is given as follows.

$$T^{i} = \left(\frac{\sum_{1}^{m} T_{m}^{i}}{m}\right) + T_{AIFS}^{i} + T_{backoff}^{i}$$
(3)

Finally, the CWT is defined as the ratio between the average duration in time the channels were found busy and the maximum interval duration. CWT is calculated using the following equation and normalized to a value between 0 and 1, i.e.,  $CWT \in [0, 1]$ .

$$CWT = \left(\frac{\sum_{i=0}^{q-1} T^i}{T_{max}}\right) \tag{4}$$

#### 2) CALCULATING QUEUE LENGTH (QL)

QL is the degree with which queues are occupied. Consider the following parameter settings.

- Maximum Capacity of all the queues:  $Q_{max} = 50$ .
- Number of neighbors: n.
- Number of messages received from the *j<sup>th</sup>* neighboring CS: *R<sub>j</sub>* where j ∈ 0, 1, 2, ... n−1.
- Number of messages sent by the current CS k:  $S_k$ .

It is assumed that each queue service the transmission and reception packets, and the system is not full-duplex. The rationale behind counting the number of packets received and sent is to signify the congestion level of the CS itself and the resources occupied by the other neighboring CS. The current size of the  $i^{th}$  queue at the  $k^{th}$  CS is based on the following equation.

$$Q_k^i = \sum_{j=0}^{n-1} R_j^i + S_k^i$$
 (5)

Finally, QL is given as the ratio between the total queue size and maximum queuing capacity using the following equation and normalized to a value between 0 and 1, i.e.,  $QL \in [0, 1]$ .

$$QL = \left(\frac{\sum_{i=0}^{q-1} Q_k^i}{Q_{max}}\right) \tag{6}$$

#### 3) CALCULATING CHANNEL OCCUPANCY LEVEL (COL)

COL is used as the criteria for selecting DSRC-enabled interface. It is calculated by combining the values of CWT and QL as follows:

$$COL = (\alpha) \times CWT + (1 - \alpha) \times QL \tag{7}$$

where  $\alpha$  is a weighting factor i.e.,  $\alpha \in [0,1]$ . The parameter  $\alpha$  is close to 1 if the CWT is more important and close to 0 when the QL is more important. More formally expressed in the following equation.

$$COL = \begin{cases} CWT, & \text{if } \alpha = 1\\ (\alpha) \times CWT + (1 - \alpha) \times QL, & \text{if } 0 < \alpha < 1 \\ QL, & \text{if } \alpha = 0 \end{cases}$$
(8)

## B. NETWORK LOAD AS LTE SELECTION CRITERIA

The basis for the selection of Secondary RAT is the knowledge about the cellular network load. As suggested in [24] and [40], the cellular LTE network load is primarily measured in terms of the number of free Physical Resource Blocks (PRBs) that are available to schedule. Fewer PRB available indicates a higher load in the connected cell and vice versa. Each CS gets the feedback regarding the LTE network load using LTE Load Indication (LLI) [24]. The 3-bit LLI value indicates eight different load states in both uplink and downlink for a particular LTE cell. The LLI can be received on the Physical Downlink Shared Channel (PDSCH).

The Call Admission Control (CAC) is one of the RRM functions in the LTE Evolved Packet System (EPS) to control and maintain the QoS of the admitted traffic flows at the bearer level. In this article, we utilized the CAC model given in [40], which rely on the partial and full preemption of application session based on the priority. Consider the following parameter settings.

- Total load contribution over a given time window: *L<sub>current</sub>*.
- Requested resources in Physical Radio Block (PRB) to achieve bit rate b: *L<sub>new</sub>(b)*.
- Total capacity of a single-macro LTE cell: C.
- Total resources needed to preempt for admitting new session:  $\Delta \rho$ .

Then, the criteria for admission control and resource allocation PRB(b) of a new session with load  $L_{new}(b)$  is given as follows:

$$L_{new}(b) = \begin{cases} PRB(b), & \text{if } L_{new}(b) \le C - L_{current} \\ PRB(b), & \text{if } L_{new}(b) \le C - (L_{current} - \bigwedge \rho) \\ 0, & otherwise \end{cases}$$
(9)

The first condition checks if sufficient resources are available for allocation with the required data rate. The second condition describes the scenario where needed resources are only made possible by either partially or fully preempt the lower priority application session. Finally, the last term represents the situation where no resources are available for allocation.

## V. PROPOSED RAT SELECTION AND VHO ALGORITHMS

The proposed protocol suite mainly consists of three algorithms (1) RAT selection, (2) VHO, and (3) DCM. The overall objective is to ensure seamless connectivity and improve performance while minimizing the number of handovers and associated latency and loss of data throughput. When a new application flow is originated, a bootstrap RAT selection procedure is performed, which selects an initial RAT based on the application type. For the adaptive RAT selection, the respective RRM entities are checked whether the application flow can be accommodated as permissible by its QoS requirements. If the measured channel occupancy level and the network load is within the pre-specified threshold value, the flow is directed to the Primary RAT and Secondary RAT, respectively. The VHO algorithm initiates the DCM procedure before making the handover decision to avoid unnecessary handover between RATs. The DCM applies beaconing rate adaptation and admission control techniques using the Primary QoS-RR Broker and Secondary QoS-RR Broker, respectively. The traffic is carried over the selected RAT until it is inevitable, and the VHO is performed.

## A. APPLICATION PROFILER AND THE RAT SELECTION ALGORITHM

Application profiler supports automatic derivation, storage, and setting of application's QoS requirements. The Application and Facilities layers define the QoS requirements on a per-application traffic flow basis. The first step in the provisioning of an application traffic flow is to specify its QoS requirements in terms of its beaconing rate. Initially, the Application layer assigns a unique identifier. It classifies application into one of the four application types, i.e., (1) Active Road Safety, (2) e-V2X, (3) Traffic Efficiency, and (4) Infotainment along with different priorities and traffic classes. For each application traffic flow, the Application Profiler in the Facilities layer manages the following QoS and system-level parameters. Table 1 summarizes the application types and initial parameter settings [37], [41], [42].

TABLE 1. Applications and traffic types with initial parameter	
settings [37], [41], [42].	

Application	Preferred	Beaconing	QCI	Priority
type	RAT	Rate		
Road safety	Primary	1-10	75	2.5 (GBR)
Traffic efficiency	Primary	1-10	79	6.5 (Non-GBR)
Infotainment	Primary	1-2	79	6.5 (Non-GBR)
e-V2X	Secondary	2-50	84	2.4 (GBR)
Voice	Secondary	0.1-10	1/7	2/7
Live Streaming	Secondary	0.1-10	2/7	4/7
Interactive	Secondary	0.1-10	3/7	3/7
Best effort	Secondary	0.1-10	6/8	6/8

- Application Identifier and type.
- Priority and traffic class, since application flows can take any of the Primary or Secondary RAT interfaces for communication, for priority assignment LTE convention are followed along with QoS Class Indicator (QCI) both in Guaranteed Bit Rate (GBR) and Non-Guaranteed Bit Rate (Non-GBR).

- Preferred or default RAT.
- QoS parameter in terms of beaconing rate, i.e., the number of beacons transmitted per second. The value varies between 1-10 packets per second, with some applications require a maximum of up to 50. The application profiler maintains three values for each application flow i.e., *minimum*, *maximum* and *tolerance*.

The application layer initiates data transmission via default RAT, selected according to the *Bootstrap RAT selection* strategy. Based on the application requirements, the Application Profiler checks the following rules to decide an appropriate access network:

- 1) If the initial RAT selection preference is given based on application type and traffic, the Secondary RAT interface for the streaming and Primary RAT interface if the application is of active road safety or traffic efficiency type.
- 2) Else if the Application Profiler checks whether the destination can be reached directly by finding an entry in the local neighbor or routing tables. Given it is true, the Primary RAT interface is selected.
- 3) Finally, the flow control is transferred to the CRRM, which initiates the RAT selection and DCM procedures by sending the *QoS Request message* either to the Primary RRM or Secondary RRM.
- 4) If the Primary RRM receives the *QoS Request message* from the CRRM, it first checks whether the application QoS requirements can be supported by the Primary RAT interface.
  - a) Primary RRM monitors different local measurements such as CWT and QL, to determine Channel Occupation Level (COL), as described in Section IV. The Primary RRM considers maximum channel occupancy, which in practice can be determined by either  $T_{max}$  or  $Q_{max}$ . The fixed threshold limit ratio over the maximum channel occupancy  $\beta \in [0,1]$  is used to calculate the effective COL i.e.,  $\omega = \beta \times T_{max}$  or  $= \beta \times Q_{max}$ . If the COL is well under the given threshold limit  $\omega$ , i.e.,  $COL < \omega$ , the CRRM is notified with a *QoS Reply Grant message*. The Primary RAT interface is selected, and the data dissemination over the selected interface is carried out.
  - b) Else, if the estimated COL value exceeds specified threshold limit value  $\omega$ , i.e.,  $COL > \omega$ , the CRRM is notified with a *QoS Reply Decline message* indicating the current channel condition is saturated that could lead to performance degradation.
  - c) The Primary RRM initiates the DCM via the Primary QoS-RR Broker procedure, as given in Section VI.
- 5) If the Secondary RRM receives the *QoS Request message* from the CRRM, then the following steps are performed.

- a) Based on the application's QoS requirements given by the application profiler, the Secondary RRM maps application traffic flow to one of the traffic classes of the selected E-UTRAN.
- b) The Secondary RRM located at the eNodeB measures the load contribution in terms of minimum or guaranteed bit data rate based on the traffic class. If the resources are enough to satisfy the application requirements, the CRRM is notified to select the Secondary RAT interface with a *QoS Reply Grant message*. The data is then forwarded over the selected interface. Else, if the measured load exceeds the capacity, i.e., LTE doesn't have enough resources, a *QoS Reply Decline message* is sent to the CRRM.
- c) Subsequently, the Secondary RRM initiates the DCM through the Secondary QoS-RR Broker procedure, as described in Section VI.

## **B. VHO DECISION ALGORITHM**

The VHO decision includes two aspects, i.e., VHO signaling and VHO algorithm. The VHO is terminal initiated and consists of the following steps.

- 6) From Primary RAT interface to Secondary RAT interface.
  - a) A CS using the Primary RAT interface monitors the channel occupation level (COL). If the COL values reach a certain threshold limit, the Primary RRM tries to reduce the channel occupancy level by performing the Primary QoS-RR Broker process.
  - b) If the Primary CRRM decides that VHO is inevitable for an application to function, it sends a request to the CRRM for handover initiation to the Secondary RRM.
  - c) The CRRM sends the handover preparation request to the Secondary RRM and queries the availability of resources for the given application.
  - d) Upon successful allocation of required resources, a VHO from DSRC to LTE is performed.
- 7) From Secondary RAT interface to Primary RAT interface.
  - a) The Secondary RRM periodically reviews all the sessions within the single macro-cell area to evaluate the offered traffic load.
  - b) If a request for a new session arrives from a vehicular user, then the Secondary RRM performs the Secondary QoS-RR Broker process by matching the requirements of the new traffic class with the available radio resources. It tries to reach an agreement with other sessions to accommodate the new application session by preempting lower priority sessions so that the overall load stays well below the total capacity.
  - c) On successfully acquiring the required resources by the Secondary RRM, the resources are

reserved, and the application flow continues to disseminate data using the Secondary RAT interface.

d) Else, for the preempted application sessions, the CRRM is notified, and a VHO is performed from LTE cellular network to DSRC.

If switching to the Primary RAT and Secondary RAT interfaces are declined, the CRRM waits for an arbitrary length of time and re-initiates the RAT selection procedure. Fig. 4 shows the signaling processes to coordinate among different elements of the proposed architecture for an efficient RAT selection, VHO, and DCM.

## VI. DYNAMIC COMMUNICATION MANAGEMENT (DCM)

While adaptive load threshold criteria based RAT selection decision results in an efficient solution, the parameter values keep going up and down, resulting in the ping-pong effect [13], i.e., frequent VHO between the Primary and Secondary RATs. The Primary and Secondary RRM entities avoid unnecessary vertical handovers by applying dynamic communication management via implementing QoS-RR Broker procedures in their respective access networks. In an attempt to reduce the channel load and thus a potential handover, the Primary QoS-RR Broker assigns an effective beaconing rate between the minimum and maximum beaconing rates as permissible by the application's QoS requirements. Similarly, the Secondary QoS-RR Broker gradually preempts low priority sessions to accommodate increasing load demand while avoiding handover to the Primary RAT.

## A. PRIMARY QoS-RR BROKER

Most of the applications in the vehicular network domain rely on detailed data obtained both from local sensors and neighboring CSs by transmitting beacons or heartbeat messages. The data provide drivers or autonomous systems an extensive field of view on the driving environment. The reliability and accuracy of vehicular applications depend on real-time beacon transmission and processing to allow the driver to react to any potentially hazardous situation appropriately within a critical time window. Higher beaconing rates, i.e., many beacons per second, can saturate the channel, causing traffic congestion, higher latency, and lower throughput [1]. On the other hand, lower beaconing rates can severely impact the accuracy of the application. Therefore, the beaconing rate can be utilized as a numerical QoS metric [43] for the proposed system. Moreover, different applications involve different beaconing rates; for example, pre-crash road safety application requires a beaconing rate of 50 beacons/sec. In contrast, community service applications can operate with as low as one beacon/sec. [5], [44]. Therefore, instead of assigning an optimal beaconing rate to all the CSs, different beaconing rates are assigned depending on the application QoS requirements, time-dependent vehicle kinetics, and the surrounding environment.

## IEEE Access



FIGURE 4. Signaling procedures for RAT selection, Vertical Handover (VHO), and Dynamic Communication Management (DCM).

In vehicular ad hoc networks, access to the wireless medium is uncoordinated; therefore, the radio resources are managed and utilized in a distributed manner. In this article, it is assumed that each CS operating the Primary RAT interface adapts the beaconing rate within a range between  $BR_{min}$ and  $BR_{max}$ . The Primary RRM negotiates a lower beaconing rate for the applications. The adaptation mechanism gradually decreases the beaconing rate in multiple steps permissible by application QoS requirements and channel conditions. The selected value is then updated and stored in the data structure maintained by the application profiler. The CSs coordinate via Information Reporting and Decision Support functions by sending and receiving QoS Request and QoS Decline control messages. The interval adaptation procedure can be initiated by periodically monitor the value of COL or triggered as the result of receiving a QoS Request message from one of the neighboring CS CRRM.

The proposed beaconing rate adaptation algorithm relies on the following parameters.

- A CS can select a beaconing rate given by  $BR_{current}$  between  $BR_{min}$  and  $BR_{max}$ , i.e.,  $BR_{min} \leq BR_{current} \leq BR_{max}$ . Initially, the current beaconing rate, is set to  $BR_{max}$ .
- Beaconing rate tolerance  $BR_{tolerance}$  given as the percentage of the  $BR_{max}$ . It is defined as the maximum tolerable reduction in beaconing rate by an application based on its QoS requirements in order to ensure the required accuracy of the application.
- Beaconing rate reduction factor *BR<sub>factor</sub>* is defined as the percentage decrease in current beaconing rate.
- $T_{initial}$  is a timer which defines the minimum time period that an application must operate at  $BR_{max}$ .
- *T<sub>reduce</sub>* is a timer which defines the maximum time duration an application can operate at the reduced beaconing rate.

The following are the steps of the beaconing rate adaption algorithm. Fig. 5 shows the flowchart diagram of the beaconing rate adaptation mechanism.



FIGURE 5. Flowchart diagram of the beaconing rate adaptation algorithm.

- 1) During the uncongested scenarios i.e.,  $COL \ll \omega$ , the applications operate at  $BR_{max}$  for minimum  $T_{initial}$  interval.
- 2) As the congestion level exceeds the given threshold value, i.e.,  $COL \ge \omega$ , the Primary QoS-RR Broker repeatedly applies the following steps for each application flows currently originating from the CS.
- 3) Gradually decrease the beaconing rate by the given  $BR_{factor}$  parameter until the beaconing rate cannot be reduced further, i.e., parameter  $BR_{tolerance}$  is reached. The proposed mechanism calculates the  $BR_{current}$  given as:

$$BR_{current} = [BR_{current} - (BR_{factor}\% \times BR_{max})] \quad (10)$$

- 4) Start a timer  $T_{reduce}$  and transmit using the current beaconing rate of  $BR_{current}$ .
- 5) The steps mentioned above are repeated for each application until the COL value becomes lower than the specified threshold limit, i.e.,  $COL < \omega$ .
- 6) The value of the current beaconing rate of  $BR_{current}$  is determined based on the following criteria.

$$BR_{current} = \begin{cases} BR_{max}, \text{ if } COL \ll \omega \\ \lceil BR_{current} - (BR_{factor} \% \times BR_{max}) \rceil, \\ \text{ if } COL \ge \omega \\ BR_{tolerance}, otherwise \end{cases}$$
(11)

#### **B. SECONDARY RR-QoS BROKER**

The Secondary RR-QoS Broker facilitates the negotiation between the applications QoS requirements and the available

resources. On selecting the Secondary RAT interface, the Secondary RRM maps the application traffic to one of the traffic classes of the cellular network.

- To accommodate requested bandwidth, the Secondary RR-QoS Broker utilizes CAC functionalities to control the LTE EPS bearers to maintain the QoS of the admitted bearers.
- 2) On finding that the load contributed can't be accommodated, the CAC applies preemption of low priority session by initiating a two-step negotiation process based on the criteria described in Section IV.
  - a) Firstly, partially preempt the resources with the lowest priority traffic class to their minimum data rate requirements.
  - b) Secondly, if the above step fails, entirely preempt the lowest priority traffic.
- 3) If the EPS bearer setup request by the Secondary RRM is accepted, the required bandwidth is allocated, and the CRRM is notified subsequently.
- 4) The CRRM registers with the cellular network and starts the VHO signaling for data communication.

### **VII. PERFORMANCE EVALUATION**

This section describes a simulation-based study of the proposed scheme. The main focus is to evaluate the impact of hybrid architecture and distributed radio resource management on the performance of the DSRC based vehicular ad hoc network. The performance is assessed in terms of vital communication and networking metrics such as the number of VHOs performed, throughput, Packet Delivery Ratio (PDR), and latency. Mainly two simulation studies are conducted.



FIGURE 6. Impact of parameters: (a) *BR<sub>factor</sub>*, (b) *BR<sub>tolerance</sub>* and (c) Combined *BR<sub>factor</sub>* and *BR<sub>tolerance</sub>* on Number of VHOs.

TABLE 2. Simulation parameters and values.

Parameters	Values		
Road length	5 km		
Lane width	5 m		
Number of lanes	3		
Number of vehicles	150		
Minimum speed	50 km/h		
Maximum speed	120 km/h		
Simulation time	300 s		
DSRC communication range	250 m		
DSRC Data rate	6 Mbps		
α	0.5		
Threshold limit ratio $\beta$	80%		
Beaconing rate	1 to 20 beacons/s		
Number of LTE cells (eNodeB)	1		
LTE bandwidth (Uplink/Downlink)	50 Mbps/50 Mbps		
LTE time slot	1 ms		
LTE resource blocks	6		
LTE total number of subcarriers	72 (12 x 6 resource blocks)		

- 1) Study the impact of different parameters on the performance.
- 2) A comparative study with several simulation scenarios such as periodic RAT switching, DSRC only, proposed hybrid approach and proposed a hybrid approach with DCM.

### A. SIMULATION ENVIRONMENT

The proposed protocol performance is validated through extensive simulations using the Matlab tool. The simulation environment represents a typical highway scenario of a 5 km road strip. The highway layout consists of three lanes on both sides, where each lane is 5 m wide. A total of 150 vehicles participated in the simulation, each traveling with a varying speed between 50 km/h to 120 km/h. The ad hoc network part is simulated using the DSRC compliant standard with an application, EDCA-based MAC, and PHY layers. The application layer is responsible for generating CAM/DENM messages at a different beaconing rate ranging between 1 beacon/sec. to 20 beacons/sec. The maximum beaconing rate is set to 20 beacons/sec. because some of the advanced V2X applications pre-crash warning require the rate of up to 50 beacons/sec. [42], [44]. Each CS periodically transmits beacons of 250 bytes size using a single-hop broadcast. The ad hoc interface operated at 5.9 GHz frequency with a 6 Mbps data rate. The maximum communication range is set to 250 m. A two-ray ground reflection model is included in the PHY layer. Generally, the highway scenario has fewer obstructions, and most vehicles have a Line-of-Sight (LOS) link among them for longer durations [45].

For modeling cellular networks, we used a simplified setup with a single cell, i.e., the radio access network, which constitutes only one eNodeB. The focus is on modeling the uplink or Uu radio interface between the vehicle and radio access network (RAN). All CS communicates through LTE broadcast on the downlink and unicast in the uplink using the Secondary RAT interface. We used the Friis channel model for cellular communication. The simplified channel models provided a reasonable abstraction level at the PHY layer and a better trade-off between the simulation complexity and details. Table 2 summarizes the stimulation parameter settings for both DSRC and cellular technologies. The results obtained were averaged over ten different simulation runs.

### **B. SIMULATION RESULTS**

1) STUDY THE IMPACT OF PARAMETERS ON PERFORMANCE Fig. 6 represents the relationship between two critical parameters, i.e.,  $BR_{tolerance}$  and  $BR_{factor}$ . Fig. 6 (a) shows that the number of VHO decreases as the  $BR_{factor}$  parameter value



FIGURE 7. Impact of parameters: (a) Treduce, (b) Tinitial and (c) Combined Treduce and Tinitial on Number of VHOs.

increases, as long as the maximum tolerable reduction is high. In such applications, a higher BR<sub>factor</sub> values impact the number of VHOs, which drops considerably. Scenarios where the maximum tolerable reduction  $BR_{tolerance}$  is negligibly low results in a higher number of VHOs. These scenarios mostly arise in applications like pre-crash sensing, remote automated driving, which are more sensitive to reducing the beaconing rate. For applications less acceptable to the decrease in beaconing rate, even decreasing the beaconing rate by 25% results in a high number of VHOs. The obtained results are mainly because the parameter  $BR_{tolerance}$  threshold is attained very quickly in the first couple of iterations that wouldn't allow the beaconing adaptation mechanism to be fully applied. These applications tend to switch more often to maintain the pre-specified QoS requirements, thus causing a higher number of VHOs. Fig. 6 (b) shows that the higher the applications can tolerate the reduction in its beaconing rate requirements, the lower the number of VHOs it performed. The higher  $BR_{tolerance}$  threshold value allows applications to operate at lower beaconing rates without sacrificing their QoS requirements, which translates into lower network load and, therefore, fewer VHOs. Fig. 6 (c) shows the combined impact of BRtolerance threshold and BRfactor parameters on the number of VHOs. Unless the application is more flexible towards reduced beaconing rate, i.e., BRtolerance value considerably more significant than  $BR_{factor}$ , the impact of higher BR<sub>factor</sub> values on the number of VHOs performed remains less signification. There is a sharp decline in the number of VHOs as soon as the BRtolerance threshold values exceed the BR<sub>factor</sub> values.

Fig. 7 shows the impact of two timer values on the number of VHOs, i.e.,  $T_{reduce}$  and  $T_{initial}$ . Fig. 7 (a) illustrates that the longer an application operates with reduced beaconing

rate, i.e., higher  $T_{reduce}$ , the lower the number of VHOs. For applications that are less sensitive to lower beaconing rates, there are lesser chances of channel occupancy to grow beyond the given threshold limit, therefore resulting in fewer VHOs. Fig. 7 (b) shows that the longer an application maintains the initial beaconing rate, i.e., T<sub>initial</sub> duration, the higher the number of VHOs. Since the applications operate at the initial beaconing rate for an extended period, it results in considerable higher channel occupancy. The beaconing rate adaptation mechanism intervenes more often and quickly reaches the threshold limits, thus causing frequent VHOs. Fig. 7 (c) shows, as the  $T_{initial}$  increases, for all the  $T_{reduce}$ durations the number of VHOs increases. Regardless of the duration, an application operates at a lower beaconing rate. If it requires to keep the higher beaconing rate longer, this will increase the channel occupancy to the level where VHO cannot be avoided.

Similarly, as the timer  $T_{reduce}$  value increases, for all the values of timer  $T_{initial}$ , the number of VHO decreases. Shorter an application can tolerate staying at the original beaconing rate, fewer the chances that the channel occupancy to increase beyond the specified threshold limit. Therefore, fewer numbers of VHOs are required to satisfy the application's QoS requirements.

Fig. 8 (a), (b) and (c) quantifies the impact of  $BR_{tolerance}$  and  $BR_{factor}$  parameters on the throughput. Situations where a selected RAT could not further support the transmission often results in switching to the other access technology. The higher throughput is obtained by steering traffic between interfaces whenever necessary, therefore attaining better reliability. It is noteworthy the contribution of each access technology towards achievable throughput. The fewer number of VHOs can easily be translated into more traffic over

## IEEE Access



FIGURE 8. Impact of parameters: (a) BR<sub>factor</sub>, (b) BR<sub>tolerance</sub> and (c) Combined BR<sub>factor</sub> and BR<sub>tolerance</sub> on Throughput (kbps).



FIGURE 9. Impact of parameters: (a) T<sub>reduce</sub>, (b) T<sub>initial</sub> and (c) Combined T<sub>reduce</sub> and T<sub>initial</sub> on Throughput (kbps).

the DSRC interface. Conversely, a higher number of VHOs resulted in more traffic pass through the LTE interface. As the values of  $BR_{factor}$  and  $BR_{tolerance}$  increases, there were fewer VHOs, which causes most of the traffic to go over the direct DSRC enabled links. The throughput decreases slightly as the  $BR_{tolerance}$  threshold increases. Despite that the combined impact of  $BR_{tolerance}$  and  $BR_{factor}$  result in lower throughput, the achieved reliability of over 92% satisfies the reliability requirements for most vehicular networking applications. Fig. 9 (a), (b) and (c) quantifies the impact of  $T_{reduce}$  and  $T_{initial}$  parameters on throughput. The higher values of  $T_{reduce}$  results in a fewer number of VHOs, which results in more

traffic to go through over the Primary RAT interface. As the value of  $T_{initial}$  increases, there was a higher number of VHOs, thus more traffic over the Secondary RAT interface resulting in higher reliability. The combined impact of  $T_{reduce}$  and  $T_{initial}$  shows that the proposed scheme can attain better reliability and throughput if the applications operate at lower beaconing rates for longer durations.

Fig. 10 (a), (b) and (c) quantifies the impact of  $BR_{factor}$  and  $BR_{tolerance}$  parameters on the latency. As the values of  $BR_{factor}$  increases, the lower number of VHOs causes vehicles to transmit more data over the Primary RAT interface. Although direct communications using the Primary RAT interface are



FIGURE 10. Impact of parameters: (a) BRfactor, (b) BRtolerance and (c) Combined BRfactor and BRtolerance on latency.



FIGURE 11. Impact of parameters: (a) Treducer (b) Tinitial and (c) Combined Treduce and Tinitial on latency.

generally faster, the latency increases due to the delays caused by the presence of hidden nodes, contention, and retransmissions, as shown in Fig. 10 (a). Additionally, the RAT selection and switching latency caused by the network selection delay is also considerable. The beaconing rate adaptation and DCM mechanisms can effectively reduce the latency if the application QoS requirements permit to operate at lower beaconing rates. Fig. 10 (b) and (c) show that the increase in the value of  $BR_{tolerance}$  causes the delay to decrease on average by 11%. Fig. 11 (a), (b) and (c) quantifies the impact of  $T_{reduce}$  and  $T_{initial}$  parameters on the latency. The higher values of  $T_{reduce}$ results in longer duration vehicles need to operate in reduced beaconing rates, which results in lower latency. The increase in the value of  $T_{initial}$  results in higher latency. The proposed scheme can reduce the latency; however, if the applications demand longer  $T_{initial}$  durations, then the latency increases.

2) COMPARATIVE STUDY AMONG DIFFERENT APPROACHES Fig. 12 shows the comparison among different schemes, including (1) periodic RAT selection and VHO, where the vehicles switch every pre-specified length of time, (2) DSRC only, (3) proposed hybrid scheme, and (4) proposed hybrid scheme with DCM. For these simulations, the average values of major simulation parameters are selected. The values set for the parameters  $BR_{factor}$ ,  $BR_{tolerance}$ ,  $T_{reduce}$ , and  $T_{initial}$  are 50%, 75%, 10 sec., and 10 sec., respectively.



FIGURE 12. Comparison among different approaches, (1) Periodic VHO, (2) DSRC only, (3) Proposed Hybrid, and (4) Proposed Hybrid with DCM: (a) Number of VHOs. (b) Latency (ms). (c) Packet Delivery Ratio (%). (d) Throughput (kbps).

Fig. 12 (a) shows the number of VHOs performed by each scheme. For the periodic switching between the two access technologies, a higher number of VHOs are reported for the lower periodicity interval. The number of VHOs in the simple hybrid schemes is quite comparable with the periodic scheme with the switching interval set to 10s. The graph illustrates the minimum, median, and maximum number of VHOs for the proposed hybrid scheme with DCM. The minimum number of VHOs is significantly lower than any other approach, signifying less reliance on the LTE interface while reducing VHO cost in terms of latency and the corresponding loss in data throughput. As shown in Fig. 12 (b), the DSRC only approach results in considerable delays. Despite the direct communication links among the vehicles, the excessive amount of traffic through a single wireless interface often resulted in a severe congestion scenario, and therefore latency increases. The RAT selection and VHO schemes incur latency, which is caused by the network discovery and selection delays and processing latency to reach the switching decision. Switching RAT intelligently when it is barely necessary, the hybrid approach with the DCM scheme resulted in a lower delay.

In the simple hybrid approach, the higher number of VHOs resulted contributed towards a higher latency cost. For the periodic switching schemes, a higher number of VHOs resulted in considerably higher latency. Fig. 12 (c) compares performance in terms of PDR. The PDR for the periodic switching approach varies between 80% to 90%, whereas the PDR for the hybrid approaches are quite comparable and reaches around 95%. The DSRC only approach suffers from severe reliability issues, especially in high load scenarios. Fig. 12 (d) compares different approaches in terms of

VOLUME 8, 2020

throughput. Hybrid approaches attain comparable throughput. However, the amount of data transferred through each access technology differs. As for the periodic switching, both interfaces transferred an equal amount of data. The hybrid-only approach performed well, however, at the cost of significantly higher delays with more data transferred over the LTE interface. In the proposed hybrid schemes, the DSRC interface delivers most of the data, whereas the use of the LTE interface is reasonably lower than all other schemes. For DSRC lower PDR, is translated in lower throughput attained.

#### **VIII. CONCLUSION**

This article proposed architecture and a suite of protocol for DSRC and C-V2X integrated hybrid vehicular networks. We address the problem of RAT selection, VHO, and data dissemination in a highway environment. The protocol suite includes an enhanced network protocol stack, an adaptive RAT selection, and VHO algorithm and dynamic communication management (DCM) algorithms. The main features of the proposed solution are: (1) A distributed, two-tiered CellCar Radio Resource Management (CRRM) module that manages the radio resources in the Primary RAT, i.e., DSRC and provides coordination with the Secondary RAT, i.e., LTE. (2) An adaptive RAT selection algorithm that takes in to account QoS metrics such as COL in Primary RAT and network load in Secondary RAT to select the most appropriate RAT according to network conditions. (3) An efficient VHO mechanism. (4) DCM via QoS negotiation by separately implementing beaconing rate adaptation and admission control in Primary RRM and Secondary RRM, respectively. The effectiveness of the proposed architecture and protocols is

evaluated using extensive simulation-based studies. In particular, several simulations were conducted with different parameter settings to get insight into the overall performance of the vehicular network in terms of the number of VHOs performed, delay, PDR, and throughput. Furthermore, a comparative study is also included, which concludes that employing a hybrid approach results in a fewer number of VHOs, higher reliability, and lower delays.

#### REFERENCES

- Z. Hameed Mir and F. Filali, "LTE and IEEE 802.11p for vehicular networking: A performance evaluation," *EURASIP J. Wireless Commun. Netw.*, vol. 2014, no. 1, pp. 1–15, May 2014, doi: 10.1186/1687-1499-2014-89.
- [2] 5G Automotive Association (5GAA). (Feb. 2019). Cellular V2X Conclusions Based on Evaluation of Available Architectural Options, White Paper. [Online]. Available: https://5gaa.org/wpcontent/uploads/2019/02/5GAA\_White\_Paper\_on\_C-V2X\_Conclusions \_based\_on\_Evaluation\_of\_Available\_Architectural\_Options.pdf
- [3] K. Z. Ghafoor, M. Guizani, L. Kong, H. S. Maghdid, and K. F. Jasim, "Enabling efficient coexistence of DSRC and C-V2X in vehicular networks," *IEEE Wireless Commun.*, vol. 27, no. 2, pp. 134–140, Apr. 2020, doi: 10.1109/MWC.001.1900219.
- [4] G. Naik, B. Choudhury, and J.-M. Park, "IEEE 802.11bd & 5G NR V2X: Evolution of radio access technologies for V2X communications," *IEEE Access*, vol. 7, pp. 70169–70184, 2019, doi: 10.1109/ACCESS.2019.2919489.
- [5] Z. H. Mir and F. Filali, "Applications, requirements, and design guidelines for multi-tiered vehicular network architecture," in *Proc. Wireless Days* (WD), Apr. 2018, pp. 15–20, doi: 10.1109/WD.2018.8361686.
- [6] Z. Hameed Mir, Y.-B. Ko, F. Filali, and J. Kim, "Infrastructure-assisted joint power adaptation and routing for heterogeneous vehicular networks," *Comput. Electr. Eng.*, vol. 73, pp. 304–320, Jan. 2019, doi: 10.1016/j.compeleceng.2018.12.005.
- [7] J. A. Olivera, I. Cortazar, C. Pinart, A. Los Santos, and I. Lequerica, "VANBA: A simple handover mechanism for transparent, always-on V2 V communications," in *Proc. IEEE 69th Veh. Technol. Conf. VTC Spring*, Apr. 2009, pp. 1–5, doi: 10.1109/VETECS.2009.5073905.
- [8] A. Hasib and A. O. Fapojuwo, "Analysis of common radio resource management scheme for end-to-end QoS support in multiservice heterogeneous wireless networks," *IEEE Trans. Veh. Technol.*, vol. 57, no. 4, pp. 2426–2439, Jul. 2008, doi: 10.1109/TVT.2007.912326.
- [9] A. M. Vegni and T. D. C. Little, "Hybrid vehicular communications based on V2V-V2I protocol switching," *Int. J. Vehicle Inf. Commun. Syst.*, vol. 2, nos. 3–4, pp. 213–231, Dec. 2011, doi: 10.1504/IJVICS.2011.044263.
- [10] K. Shafiee, A. Attar, and V. C. M. Leung, "Optimal distributed vertical handoff strategies in vehicular heterogeneous networks," *IEEE J. Sel. Areas Commun.*, vol. 29, no. 3, pp. 534–544, Mar. 2011, doi: 10.1109/JSAC.2011.110304.
- [11] K. Shafiee and V. C. Leung, "Hybrid multi-technology routing in heterogeneous vehicular networks," *EURASIP J. Wireless Commun. Netw.*, vol. 2012, no. 1, pp. 1499–1687, Dec. 2012, doi: 10.1186/1687-1499-2012-35.
- [12] Q. Zhao, Y. Zhu, C. Chen, H. Zhu, and B. Li, "When 3G meets VANET: 3G-assisted data delivery in VANETs," *IEEE Sensors J.*, vol. 13, no. 10, pp. 3575–3584, Oct. 2013, doi: 10.1109/JSEN.2013.2265304.
- [13] L. Wu and K. Sandrasegaran, A Study on Radio Access Technology Selection Algorithms (SpringerBriefs in Electrical and Computer Engineering). Berlin, Germany: Springer, 2012, doi: 10.1007/978-3-642-29399-3.
- [14] IEEE Standard for Information Technology–Telecommunications and Information Exchange Between Systems Local and Metropolitan Area Networks–Specific Requirements–Part 11: Wireless LAN Medium Access Control (MAC) and Physical Layer (PHY) Specifications, IEEE Standard 802.11-2016 (Revision of IEEE Std 802.11-2012), Dec. 2016, pp. 1–3534, doi: 10.1109/IEEESTD.2016.7786995.
- [15] LTE. (2010). Overview of 3GPP Release 8. [Online]. Available: http://www.3gpp.org/Release-8
- [16] S. Zeadally, M. A. Javed, and E. B. Hamida, "Vehicular communications for ITS: Standardization and challenges," *IEEE Commun. Standards Mag.*, vol. 4, no. 1, pp. 11–17, Mar. 2020, doi: 10.1109/MCOM-STD.001.1900044.

- [17] Y. Zhou, H. Liu, Z. Pan, L. Tian, and J. Shi, "Energy-efficient two-stage cooperative multicast: Effect of user density," *IEEE Trans. Veh. Technol.*, vol. 65, no. 9, pp. 7297–7307, Sep. 2016, doi: 10.1109/TVT.2015.2494066.
- [18] Y. Zhou, H. Liu, Z. Pan, L. Tian, and J. Shi, "Cooperative multicast with location aware distributed mobile relay selection: Performance analysis and optimized design," *IEEE Trans. Veh. Technol.*, vol. 66, no. 9, pp. 8291–8302, Sep. 2017, doi: 10.1109/TVT.2017.2682903.
- [19] J. Cheng, J. Cheng, M. Zhou, F. Liu, S. Gao, and C. Liu, "Routing in Internet of vehicles: A review," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 5, pp. 2339–2352, Oct. 2015, doi: 10.1109/TITS.2015.2423667.
- [20] IEEE Standard for Local and Metropolitan Area Networks—Part 21: Media Independent Services Framework, IEEE Standard 802.21-2017 Revision of IEEE Standard 802.21-2008 as amended by IEEE Standard 802.21a-2012, IEEE Standard 802.21b-2012, IEEE Standard 802.21c-2014, and IEEE Standard 802.21d-2015, Apr. 2017, pp. 1–314, doi: 10.1109/IEEESTD.2017.7919341.
- [21] Intelligent Transport Systems–Communications Access for Land Mobiles (CALM)—Architecture, ISO Standard 21217:2014, 2014.
- [22] J. M. Marquez-Barja, H. Ahmadi, S. M. Tornell, C. T. Calafate, J.-C. Cano, P. Manzoni, and L. A. DaSilva, "Breaking the vehicular wireless communications barriers: Vertical handover techniques for heterogeneous networks," *IEEE Trans. Veh. Technol.*, vol. 64, no. 12, pp. 5878–5890, Dec. 2015, doi: 10.1109/TVT.2014.2386911.
- [23] Z. H. Mir, J. Toutouh, F. Filali, and E. Alba, "QoS-aware radio access technology (RAT) selection in hybrid vehicular networks," in *Communication Technologies for Vehicles Nets4Cars/Nets4Trains/Nets4Aircraft* 2015 (Lecture Notes in Computer Science), vol. 9066. Cham, Switzerland: Springer, 2015, pp. 117–128, doi: 10.1007/978-3-319-17765-6\_11.
- [24] N. Dreyer, A. Moller, Z. H. Mir, F. Filali, and T. Kurner, "A data traffic steering algorithm for IEEE 802.11p/LTE hybrid vehicular networks," in *Proc. IEEE 84th Veh. Technol. Conf. (VTC-Fall)*, Sep. 2016, pp. 1–6, doi: 10.1109/VTCFall.2016.7880850.
- [25] M. Ben Brahim, Z. Hameed Mir, W. Znaidi, F. Filali, and N. Hamdi, "QoS-aware video transmission over hybrid wireless network for connected vehicles," *IEEE Access*, vol. 5, pp. 8313–8323, 2017, doi: 10.1109/ACCESS.2017.2682278.
- [26] Z. H. Mir, J. Kim, F. Filali, and Y.-B. Ko, "Infrastructure-assisted efficient broadcasting in hybrid vehicular networks," in *Proc. Int. Conf. Connected Vehicles Expo (ICCVE)*, Oct. 2015, pp. 248–253, doi: 10.1109/ICCVE.2015.29.
- [27] G. Remy, S. Senouci, F. Jan, and Y. Gourhant, "LTE4 V2X: LTE for a centralized VANET organization," in *Proc. IEEE Global Telecommun. Conf. (GLOBECOM)*, Dec. 2011, pp. 1–6, doi: 10.1109/GLO-COM.2011.6133884.
- [28] H.-Y. Pa, R.-H. Jan, A. A.-K. Jeng, C.-C. Chen, and H.-R. Tseng, "Mobile gateway routing for vehicular networks," in *Proc. 8th IEEE Asia Pacific Wireless Commun. Symp.*, Aug. 2011, pp. 1–5.
- [29] A. Benslimane, T. Taleb, and R. Sivaraj, "Dynamic clustering-based adaptive mobile gateway management in integrated VANET—3G heterogeneous wireless networks," *IEEE J. Sel. Areas Commun.*, vol. 29, no. 3, pp. 559–570, Mar. 2011, doi: 10.1109/JSAC.2011.110306.
- [30] R. Sivaraj, A. K. Gopalakrishna, M. G. Chandra, and P. Balamuralidhar, "QoS-enabled group communication in integrated VANET-LTE heterogeneous wireless networks," in *Proc. IEEE 7th Int. Conf. Wireless Mobile Comput., Netw. Commun. (WiMob)*, Oct. 2011, pp. 17–24, doi: 10.1109/WiMOB.2011.6085417.
- [31] G. E. M. Zhioua, H. Labiod, N. Tabbane, and S. Tabbane, "An efficient qos based gateway selection algorithm for VANET to LTE advanced hybrid cellular network," in *Proc. 15th ACM Int. Conf. Modeling, Anal. Simulation Wireless Mobile Syst. (MSWiM)*, 2012, pp. 353–356, doi: 10.1145/2387238.2387297.
- [32] G. el Mouna Zhioua, H. Labiod, N. Tabbane, and S. Tabbane, "FQGwS: A gateway selection algorithm in a hybrid clustered VANET LTE-advanced network: Complexity and performances," in *Proc. Int. Conf. Comput., Netw. Commun. (ICNC)*, Feb. 2014, pp. 413–417, doi: 10.1109/ICCNC.2014.6785370.
- [33] I. Turcanu, F. Klingler, C. Sommer, A. Baiocchi, and F. Dressler, "Duplicate suppression for efficient floating car data collection in heterogeneous LTE-DSRC vehicular networks," *Comput. Commun.*, vol. 123, pp. 54–64, Jun. 2018, doi: 10.1016/j.comcom.2018.03.015.
- [34] CAR 2 CAR Communication Consortium. Accessed: Aug. 31, 2020. [Online]. Available: https://www.car-2-car.org/

- [35] Intelligent Transport Systems (ITS); Vehicular Communications; Basic Set of Applications; Part 2: Specification of Cooperative Awareness Basic Service, document ETSI TC ITS EN 302 637-2, 2014.
- [36] Intelligent Transport Systems (ITS); Vehicular Communications; GeoNetworking; Part 3: Network Architecture, document ETSI EN 302 636-3, 2014.
- [37] 3rd Generation Partnership Project; Technical Specification Group Services and System Aspects; Study on Enhancement of 3GPP Support for 5G V2X Services (v16.2.0, Release 16), document 3GPP TR 22.886, 3GPP, Dec. 2018.
- [38] D. Puthal, Z. H. Mir, F. Filali, and H. Menouar, "Cross-layer architecture for congestion control in vehicular ad-hoc networks," in *Proc. Int. Conf. Connected Vehicles Expo (ICCVE)*, Dec. 2013, pp. 887–892, doi: 10.1109/ICCVE.2013.6799921.
- [39] M. A. Labiod, M. Gharbi, F.-X. Coudoux, P. Corlay, and N. Doghmane, "Enhanced adaptive cross-layer scheme for low latency HEVC streaming over vehicular ad-hoc networks (VANETs)," *Veh. Commun.*, vol. 15, pp. 28–39, Jan. 2019, doi: 10.1016/j.vehcom.2018.11.004.
- [40] M. Khabazian, O. Kubbar, and H. Hassanein, "A fairness-based preemption algorithm for LTE-advanced," in *Proc. IEEE Global Commun. Conf. (GLOBECOM)*, Dec. 2012, pp. 5320–5325, doi: 10.1109/GLO-COM.2012.6503966.
- [41] 3rd Generation Partnership Project; Technical Specification Group Services and System Aspects; Policy and Charging Control Architecture, (Release 16), document 3GPP TS 23.203 V16.2.0 (2019-12-22), 2019.
- [42] G. Pocovi, M. Lauridsen, B. Soret, K. I. Pedersen, and P. Mogensen, "Automation for on-road vehicles: Use cases and requirements for radio design," in *Proc. IEEE 82nd Veh. Technol. Conf. (VTC-Fall)*, Sep. 2015, pp. 1–5, doi: 10.1109/VTCFall.2015.7390848.
- [43] J. Toutouh and E. Alba, "A swarm algorithm for collaborative traffic in vehicular networks," *Veh. Commun.*, vol. 12, pp. 127–137, Apr. 2018, doi: 10.1016/j.vehcom.2018.04.003.
- [44] M. Boban, A. Kousaridas, K. Manolakis, J. Eichinger, and W. Xu, "Connected roads of the future: Use cases, requirements, and design considerations for vehicle-to-everything communications," *IEEE Veh. Technol. Mag.*, vol. 13, no. 3, pp. 110–123, Sep. 2018, doi: 10.1109/MVT.2017.2777259.
- [45] Z. Hameed Mir and F. Filali, "Large-scale simulations and performance evaluation of connected cars–A V2 V communication perspective," *Simul. Model. Pract. Theory*, vol. 73, pp. 55–71, Apr. 2017, doi: 10.1016/j.simpat.2017.01.004.



JAMAL TOUTOUH received the dual M.Sc. degrees from the University of Malaga, Spain, and the University of Luxembourg, Luxembourg, and the Ph.D. degree in computer engineering from the University of Malaga. He is currently a Marie Curie Postdoctoral Fellow with the Massachusetts Institute of Technology (MIT) as part of the Anyscale Learning For All (ALFA) Group. His current research interests include the combination of gradient-free (mainly

nature-inspired computing) and gradient-based optimization methods to address adversarial and generative machine learning. Besides, he works on applied machine and deep learning to smart mobility, smart cities, and climate change. His dissertation was awarded the 2018 Best Spanish Ph.D. Thesis in Smart Cities Award and the Best Ph.D. Thesis Award (University Chair Aytos-Berger Levrault on the Development of Smart Governance).



**FETHI FILALI** (Senior Member, IEEE) received the Ph.D. degree in computer science and the Habilitation degree from the University of Nice Sophia Antipolis, France, in 2002 and 2008, respectively. He was with the Mobile Communications Department, EURECOM, France, as an Assistant Professor and then an Associate Professor for eight years. He is currently the Head of Technology Development and Applied Research with the Qatar Mobility Innovations

Center (QMIC), Qatar University. He is also leading the technology development of QMIC's solutions in the areas of smart cities, the Internet of Things, intelligent transportation systems, and connected and automated vehicles solutions. He has invented technologies and developed algorithms that have been shipped in many QMIC products, including Masarak, Labeeb, and WaveTraf, creating commercial impact in the order of millions of dollars. His research grants include 15 competitive awards from several funding agencies, including the European Commission, the French National Research Agency, and the Qatar National Research Fund. He was the Ph.D. Director for over ten Ph.D. students in the areas of intelligent transportation, wireless sensor and mesh networks, vehicular communications, big data analytics, the Internet of Things, and mobility management. He has coauthored over 130 research papers in international peer-reviewed conferences and journals. He holds over ten patent applications.



**ZEESHAN HAMEED MIR** (Senior Member, IEEE) received the B.S. degree from the Sir Syed University of Engineering and Technology (SSUET), Pakistan, in 1999, the M.S. degree from the National University of Sciences and Technology (NUST), Pakistan, in 2004, and the Ph.D. degree from Ajou University, South Korea, in 2009. From 2013 to 2016, he was with the Qatar Mobility Innovations Center (QMIC), Qatar University, Qatar, as a Research Scientist. From

2009 to 2012, he was a member of Technical Staff with the Electronics and Telecommunications Research Institute, South Korea. From 2001 to 2005, he was with the Faculty of CS Department, Institute of Business Administration, Pakistan. He is currently an Assistant Professor and the Program Team Leader (PTL) with the Department of Computer and Information Science (CIS), Higher Colleges of Technology (HCT), United Arab Emirates. He has authored his research work in the significant research publications worldwide and served on the program/reviewer committees for several reputed conferences and journals. His research interests include mobile/ubiquitous computing, wireless networking/communications, smart mobility, and urban analytics.



**YOUNG-BAE KO** received the Ph.D. degree in computer science from Texas A&M University, College Station, TX, USA, in 2000. He is currently a Professor with the Department of AI Convergence Network and the Department of Computer Engineering, Ajou University, South Korea. He is also the Head of the Next-Generation Hyper-Intelligence Network Convergence Research Group, funded by the Brain Korea 21—Fostering Outstanding Universities for

Research (BK21 FOUR) national project. Prior to joining Ajou University, in 2002, he was with the IBM Thomas J. Watson Research Center, NY, USA, as a Research Staff Member with the Department of Ubiquitous Networking and Security. His recent research interests include the intelligent Internet of Things (IIoT), AI-empowered wireless communications and networking, flying/vehicular *ad hoc* networks (FANET/VANET), and super-high precision positioning technologies. He has been serving in multiple professional activities, most notably as the General Chair of the IEEE SECON 2012.

. . . .